Author's response to reviews

Title: Interaction between polymorphisms of the Human Leukocyte Antigen and HPV-16 Variants on the risk of invasive cervical cancer

Authors:

Patricia S de Araujo Souza (psas@inca.gov.br)
Paulo C Maciag (maciag@advaxis.com)
Karina B Ribeiro (kbribeiro@gmail.com)
Maria Luiza Petzl-Erler (perler@ufpr.br)
Eduardo L Franco (eduardo.franco@mcgill.ca)
Luisa L Villa (lvilla@ludwig.org.br)

Version: 4 Date: 15 August 2008

Author's response to reviews: see over
In response to your requests for formatting changes in the manuscript MS:2964239731882454, I enclose herewith a revised version of an original manuscript entitled “Interaction between polymorphisms of the Human Leukocyte Antigen and HPV-16 variants on the risk of invasive cervical cancer”. Regarding the major changes requested I provide a list of answers below. I also would like to request the journal to do the minor changes in our behalf.

1. The tables were formatted to a portrait page layout.
2. A written informed consent was obtained from the patients involved in this project for publication of this case report. A copy of the written consent (in Portuguese) is available for review by the Editor-in-Chief of this journal.
3. We modified the statement with the information about the appropriate ethics committees that approved the study, and included the bodies which gave approval: ethical review committee of the Hospital Napoleão Laureano and ethical review committee of the Hospital do Cancer A C Camargo/ Fundação Antonio Prudente Hospital do Câncer (protocol number 201/99).

Thank you for your attention,

Patricia S de Araujo Souza